Author's response to reviews

Title: Interventricular septum hematoma during cineventriculography

Authors:

Andrea Grohmann (andrea.grohmann@charite.de)
Thomas Elgeti (thomas.elgeti@charite.de)
Stephan Eddicks (stephan.eddicks@charite.de)
Fabian Knebel (fabian.knebel@charite.de)
Wolfgang Rutsch (wolfgang.rutsch@charite.de)
Christoph Melzer (christoph.melzer@charite.de)
Marko Boehm (marko.boehm@charite.de)
Gert Baumann (gert.baumann@charite.de)
Adrian C Borges (adrian.borges@charite.de)

Version: 5 Date: 10 January 2008

Author's response to reviews:

The requested changes were performed according to today's email.
Yours sincerely,
Fabian Knebel